Author's response to reviews

Title: Delayed surgical treatment for a traumatic bilateral cervical facet joint dislocation using a posterior-anterior approach. A case report.

Authors:

Takashi Shimada (shimadatakashi1204@yahoo.co.jp)
Seiji Ohtori (sohtori@faculty.chiba-u.jp)
Gen Inoue (ginoue@kitasato-u.ac.jp)
Junichi Nakamura (njonedr@yahoo.co.jp)
Izumi Nakada (izu-mel@umin.ac.jp)
Hiroshi Saiki (forza_hiro2002@yahoo.co.jp)
Song Ho Chang (s5chang@yahoo.co.jp)
Kou Kawamura (sohtori@faculty.chiba-u.jp)
Kazuhisa Takahashi (19501114@faculty.chiba-u.jp)
Hiroshi Sugiyama (hirosg1326@gmail.com)

Version: 5 Date: 15 November 2012

Author's response to reviews: see over
The Abstract does need some work. The first sentence is too brief and does not describe what the case report is about. In the Case presentation section, the authors need to remove the details about ethics approval and consent.

We thank and agree with the reviewers for the suggestions they made to improve our manuscript. We addressed abstract as the reviewer suggested.